Author’s response to reviews

Title: Total hip arthroplasty for Crowe IV hip without subtrochanteric shortening osteotomy -a long term follow up study-

Authors:

Toshiyuki Kawai (t.cawai@gmail.com)
Chiaki Tanaka (c.tanaka@kyoto.zaq.jp)
Hiroshi Kanoe (kanoe@tb3.so-net.ne.jp)

Version: 2 Date: 19 October 2012

Author’s response to reviews:

We revised the manuscript as described below.

Line 108
We added the text that describes that this study was performed with the approval of an ethics committee and was in compliance with the Helsinki Declaration.

Line 246 to 247
Competing interest section was added.

Line 249 to 252
Authors’ contributions were added.

Line 254 to 255
Acknowledgements was added.

Sincerely,
Toshiyuki Kawai